Author's response to reviews

Title: Intravitreal cilium associated with retinal detachment 40 years following penetrating eye injury: a case report

Authors:

Maria Dettoraki (mariadetor@hotmail.com)
Konstantinos Andreanos (coandre80@ioloud.com)
Stavroula Davou (st.davou@gmail.com)
Nikolaos Nomikarios (nominikos@gmail.com)
Marilitsa M Moschos (moschosmarilitsa@yahoo.fr)
Dimitrios Brouzas (brouzas@yahoo.com)

Version: 3 Date: 7 February 2015

Author's response to reviews: see over
To  
The Editorial Board of Journal of *BMC Ophthalmology*  

Athens, 07/02/2015

Dear Editor,

I am sending for publication to your distinguished Journal the revised version of the manuscript entitled: “*Intravitreal cilium associated with retinal detachment 40 years following penetrating eye injury: a case report*”.

In this revision every effort was made to meet all your requirements and the reviewers’ comments.

**Response to the reviewers’ comments**

**Reviewer: Michael Tsatsos**

“The conclusion of the manuscript needs to be changed in accordance to the manuscript theme rather than a general comment on the inert nature of intraocular cilia”

The conclusion was changed according to your suggestion.

We would like to thank you for your comments that significantly helped us to improve our manuscript and for the opportunity you give us to revise our manuscript.

We hope that this revised paper would be appropriate for publication in *BMC Ophthalmology*.

Yours sincerely,

Dimitrios Brouzas MD, PhD